Author's response to reviews

Title:Socio-cultural and behavioural factors constraining latrine adoption in rural coastal Odisha: An exploratory qualitative study

Authors:

Parimita Routray (parimita.routray@lshtm.ac.uk)
Marion W Jenkins (mwjenkins@ucdavis.edu)
Sophie Boisson (sophie.boisson@lshtm.ac.uk)
Thomas Clasen (thomas.f.clasen@emory.edu)
Wolf Peter Schmidt (wolf-peter.schmidt@lshtm.ac.uk)

Version:3 Date:29 November 2014

Author's response to reviews:

The study was approved by the Ethics Committee at the London School of Hygiene and Tropical Medicine and the Independent Ethics Committee at the Xavier Institute of Management, Bhubaneswar, India. The qualitative component of the study adheres to the RATS guidelines.

All the participants were above the age group of 16, therefore no parental consent was needed.